Author's response to reviews

Title: Implementing knowledge into practice for improved neonatal survival; a cluster-randomised, community-based trial in Quang Ninh province, Vietnam

Authors:

Lars Wallin (lars.wallin@karolinska.se)
Mats Malqvist (mats.malqvist@khh.uu.se)
Nguyen T Nga (Nga.Nguyen@khh.uu.se)
Leif Eriksson (leif.eksson@khh.uu.se)
Lars-Åke Persson (lars.ake.persson@khh.uu.se)
Dinh P Hoa (phuonghoa55@yahoo.com)
Tran Q Huy (huyub@yahoo.com)
Duong M Duc (giotamhon@gmail.com)
Uwe Ewald (uwe.ewald@khh.uu.se)

Version: 2 Date: 22 September 2011

Author’s response to reviews:

We have now made required formatting changes!